Author’s response to reviews

Title: TUCAN/CARDINAL/CARD8 and apoptosis resistance in non-small cell lung cancer cells

Authors:

Agnieszka Checinska - (a.checinska@vumc.nl)
Giuseppe Giaccone - (g.giaccone@vumc.nl)
Bas SJ Hoogeland - (hoog@mpipsyk1.mpg.de)
Carlos G Ferreira - (cferreira@inca.gov.br)
Jose A Rodriguez - (ja.rodriguez@vumc.nl)
Frank AE Kruyt - (kruyt@vumc.nl)

Version: 3 Date: 22 June 2006

Author’s response to reviews:

M Norton, MD
BMC Cancer Editorial Office

Date Reference Extension
June 22nd, 2006 ACH (+31) 20 - 4442409

Subject
Re-submission of revised manuscript

Dear Editor,

We are very happy to hear that our manuscript: "TUCAN/CARDINAL/CARD8 and apoptosis resistance in non-small cell lung cancer cells" by A.Checinska, G.Giaccone, BSJ. Hoogeland, JA. Rodriguez and FAE. Kruyt is accepted for publication in BMC Cancer.

We have made the final changes in the manuscript as requested:


2) We have converted into plain text all the changes that were previously highlighted in bold and underlined.

3) We have removed the website address (www.ncbi.nlm.nih.gov) (page 5; line 20).

Sincerely yours,

Agnieszka Checinska
Department of Medical Oncology
VU University Medical Center
a.checinska@vumc.nl